Author's response to reviews

Title: Introducing a multi-site program for early diagnosis of HIV infection among HIV exposed infants in Tanzania: Lessons from the first 6 months of implementation

Authors:

- Harriet Nuwagaba-Biribonwoha (hn2158@columbia.edu)
- Bazghina Werq-Semo (bazsemo@yahoo.co.uk)
- Aziz Abdallah (aziz_ochieng@hotmail.com)
- Amy Cunningham (ac2735@columbia.edu)
- John G Gamaliel (jgg2112@columbia.edu)
- Sevestine Mtunga (sm2759@columbia.edu)
- Victoria Nankibirwa (vn2141@columbia.edu)
- Isaya Malisa (yegeem@googlemail.com)
- Luis F Gonzalez (lg2284@columbia.edu)
- Charles Massambu (cmassambu@hotmail.com)
- Denis Nash (dn2145@mail.cumc.columbia.edu)
- Jessica Justman (eja1@mail.cumc.columbia.edu)
- Elaine J Abrams (jja2158@columbia.edu)

Version: 2 Date: 28 October 2009

Author's response to reviews:

Reviewer preferences, please see below.